Author's response to reviews

Title: Randomised placebo-controlled trial on antibiotics for bronchiectasis exacerbations in children: rationale and protocol

Authors:

Anne B Chang (annechang@ausdoctors.net)
Colin F Robertson (colin.robertson@rch.org.au)
Keith Grimwood (k.grimwood@uq.edu.au)
Peter P van Asperen (peter.vanasperen@health.nsw.gov.au)
Andrew C Wilson (Andrew.Wilson@health.wa.gov.au)
Kerry-Ann F O&#8217;Grady (k.ogrady@uq.edu.au)
Theo P Sloots (t.sloots@uq.edu.au)
Paul J Torzillo (pault@med.usyd.edu.au)
Emily J Bailey (e.bailey1@uq.edu.au)
Gabrielle B McCallum (Gabrielle.McCallum@menzies.edu.au)
Ian B Masters (brent_masters@health.qld.gov.au)
Catherine A Byrnes (c.byrnes@auckland.ac.nz)
Helen M Buntain (helen_buntain@health.qld.gov.au)
Ian M Mackay (ian.mackay@uq.edu.au)
Peter S Morris (Peter.Morris@menzies.edu.au)

Version: 3 Date: 10 May 2012

Author's response to reviews:

Dear editors,

Am attaching the ethics clearance which I had done via the email but unclear if I have to load it up here as well so have done so, as I have not a reply from the email.

Thank you again for reviewing and considering publication of our study protocol in Trials. The study is funded by Australia’s major national research funding body, the National Health and Medical Research Council (NHMRC). Our grant award letter was previously uploaded.

Thank you,

anne

anne